CST Part IB Semantics of Programming Languages List of corrections to the 99/00 lecture notes November 16, 1999

Page 41, line 18: The displayed line should read while B do ((C ; C^0) ; C^{00}) and while B do (C ; (C^0 ; C^{00})) Page 47, line -2: The last sentence of Exercise 4.2.2 should read ŞWhat happens if the locations assigned to in C are disjoint from the locations occurring in B?Ť Page 50, line -12: Replace ‘fn’ by ‘λ’.

Page 72: line 8: Replace ‘Slide 46) and’ by ‘Slide 46 and’.

Page 72, line 11: The second line in the de^nition of eq should read ‘x_1 := !x_2 + 1 andthen ’. Page 73: line 6: Replace ‘less than n’ by ‘no more than n’.